Author's response to reviews

Title: Tackling inequalities in obesity: How effective are public health interventions at reducing socio-economic inequalities in obesity amongst children?

Authors:

Clare L Bambra (clare.bambra@durham.ac.uk)
Frances C Hillier (frances.hillier@durham.ac.uk)
Helen J Moore (helen.moore@durham.ac.uk)
Carolyn D Summerbell (carolyn.summerbell@durham.ac.uk)

Version: 4 Date: 20 December 2011

Author's response to reviews: see over
Dear Editorial Team,

Please find attached the revised manuscript MS: 2377513686367679
“Tackling inequalities in obesity: How effective are public health interventions at reducing socio-economic inequalities in obesity amongst children?”.

We would like to thank the reviewer for her positive comments. We have addressed the discretionary revisions (please see below) and highlighted the edits in red in the manuscript.

1. The primary and secondary outcomes to be abstracted should be listed in the abstract, i.e. the authors should describe here how changes in economic inequalities in childhood obesity will be measured.

   Please see additional text added to the abstract (page 2)

2. Background section of the protocol is very well written but a bit long. It could be cut down to focus on the main points.

   We have removed a large amount of text throughout the background section and focused on the main points (pages 4, 5 and 6).

3. Methods section - Page 10 refers to previous systematic reviews for inclusion criteria. Would prefer a couple sentences be added to the protocol so that readers do not need to access a separate document.

   We have added more detail on the references cited (page 10).

We hope that the changes have been made to your satisfaction and we look forward to hearing from you.

Yours faithfully,

Frances Hillier

Research Associate
Department of Geography
Wolfson Research Institute
Durham University, Queen's Campus
University Boulevard
Stockton-on-Tees
TS17 6BH
Tel: +44 (0)191 3340456
Email: frances.hillier@durham.ac.uk